Author’s response to reviews

Title: The Landscape of Vaccines in China: History, Classification, Supply, and Price

Authors:
Hongjie Yu (yhj@fudan.edu.cn)

Version: 2 Date: 26 Aug 2018

Author’s response to reviews:

Dear Editor:

Here are the comments and the revisions we have made:

1. Upon submitting your revised manuscript, please ensure that all authors are added to your submission data in the same order as they are presented on the title page.

A: Yes, the order of authors has not been changed.

2. Please include the email addresses of all authors on the title page of your manuscript.

A: emails for all authors had been added in the title page.

3. Please provide a section “Additional Files” after References where you list the following information about each of your supplementary files:
- File name (e.g. Appendix)

- Title of data

- Brief description of the data

A: the descriptions has been added in the manuscript.

Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

A: checked and all additional files are explicitly referred in the main text.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A: Upload the clean version of manuscript.